

Dear Reviewers:

Thank you very much for your letter and the referees reports. Based on your comment and request, we have made modification on the original manuscript. Here, we attached revised manuscript in the formats of MS word, for your approval.

First, we added running title under the title of the article.

Second, we have submitted Non-native speakers of English editing certificate in 2019-04-19.

Third, we have submitted an audio file describing final core tip.

Fourth, we changed the format of the references. The printed copy of the first page of the full reference of references of **NO** 10 *Soft tissue tumor pathology* was submitted in the manuscript because it have not been indexed by PubMed.

Should you have any questions, please contact us without hesitate.

Sincerely yours,

Zhao-hui Liu